Author's response to reviews

Title: Hemoglobin levels and anemia evaluation during pregnancy in the highlands of Tibet: a hospital-based study

Authors:

Yuan Xing (tjxxyy@163.com)
Hong Yan (xjtu_yh.paper@yahoo.com.cn)
Shaonong Dang (tidshn@mail.xjtu.edu.cn)
Bianba Zhuoma (yliuhua@163.com)
Xiaoyan Zhou (larkluna@163.com)
Duolao Wang (duolao.wang@lshtm.ac.uk)

Version: 4 Date: 11 September 2009

Author's response to reviews: see over
11 Sep 2009

Dear Editors,

We have made formatting changes accordingly.

Major revisions:
Authors' contributions - We have indicated that all authors read and approved the final manuscript.

Minor revisions:
Line numbers - Removed
All text present in black
Abstract - The Abstract matches exactly what is in the submission system
Tracking - All tracking remove from the manuscript

With all best wishes,

Yours sincerely,

Professor Hong Yan

Professor and Dean of the College of Medicine
Deans’ Office
Xi’an Jiaotong University College of Medicine
P.O Box 46
No.76 West Yanta Road
South Suburb of Xi’an
Xi’an, 710061,
Shaanxi Province
P.R. CHINA